Practical help for specifying the target difference in sample size calculations for RCTs: the DELTA² five-stage study, including a workshop

Jonathan A Cook, 1* Steven A Julious, 2 William Sones, 1 Lisa V Hampson, 3 Catherine Hewitt, 4 Jesse A Berlin, 5 Deborah Ashby, 6 Richard Emsley, 7 Dean A Fergusson, 8 Stephen J Walters, 2 Edward CF Wilson, 9, 10 Graeme MacLennan, 11 Nigel Stallard, 12 Joanne C Rothwell, 2 Martin Bland, 13 Louise Brown, 14 Craig R Ramsay, 15 Andrew Cook, 16 David Armstrong, 17 Douglas Altman 1† and Luke D Vale 18

¹Centre for Statistics in Medicine, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK ²Medical Statistics Group, School of Health and Related Research, University of

Sheffield, Sheffield, UK

- ³Statistical Methodology and Consulting, Novartis Pharma AG, Basel, Switzerland ⁴York Trials Unit, Department of Health Sciences, University of York, York, UK ⁵Johnson & Johnson, Titusville, NJ, USA
- ⁶Imperial Clinical Trials Unit, Imperial College London, London, UK
- ⁷Department of Biostatistics and Health Informatics, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK
- ⁸Clinical Epidemiology Program, Ottawa Hospital Research Institute, Ottawa, ON, Canada
- ⁹Cambridge Centre for Health Services Research, Cambridge Clinical Trials Unit University of Cambridge, Cambridge, UK
- ¹⁰Health Economics Group, Norwich Medical School, University of East Anglia, Norwich, UK
- ¹¹Centre for Healthcare Randomised Trials, University of Aberdeen, Aberdeen, UK
- ¹²Warwick Medical School, Statistics and Epidemiology, University of Warwick, Coventry, UK
- ¹³Department of Health Sciences, University of York, York, UK
- ¹⁴MRC Clinical Trials Unit, Institute of Clinical Trials and Methodology, University College London, London, UK
- ¹⁵Health Services Research Unit, University of Aberdeen, Aberdeen, UK
- ¹⁶Wessex Institute, University of Southampton, Southampton, UK

- ¹⁷School of Population Health and Environmental Sciences, King's College London, London, UK
- ¹⁸Health Economics Group, Institute of Health & Society, Newcastle University, Newcastle upon Tyne, UK
- *Corresponding author jonathan.cook@ndorms.ox.ac.uk †In memoriam

Declared competing interests of authors: Lisa V Hampson is an employee of Novartis Pharma AG (Basel, Switzerland) and reports grants from the Medical Research Council (MRC). Catherine Hewitt is a member of the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) Commissioning Board since 2015. Jesse A Berlin is an employee of Johnson & Johnson (New Brunswick, NJ, USA) and holds shares in this company. Richard Emsley is a member of the NIHR HTA Clinical Trials Board since 2018. Deborah Ashby is a member of the HTA Commissioning Board, HTA Funding Boards Policy Group, HTA Mental Psychological and Occupational Health Methods Group, HTA Prioritisation Group and the HTA Remit and Competitiveness Group from January 2016 to December 2018. Stephen J Walters declares his department has contracts and/or research grants with the Department of Health and Social Care, NIHR, MRC and the National Institute for Health and Care Excellence. He also declares book royalties from John Wiley & Sons, Inc. (Hoboken, NJ, USA), as well as a grant from the MRC and personal fees for external examining. Louise Brown is a member of the NIHR Efficacy and Mechanism Evaluation Board since 2014. Craig R Ramsay is a member of the NIHR HTA General Board since 2017. Andrew Cook is a member of the NIHR HTA Interventional Procedures Methods Group, HTA Intellectual Property Panel, HTA Prioritisation Group, Public Health Research (PHR) Research Funding Board, Public Health Research Prioritisation Group and the PHR Programme Advisory Board.

Published October 2019 DOI: 10.3310/hta23600

Plain English summary

The DELTA² five-stage study, including a workshop

Health Technology Assessment 2019; Vol. 23: No. 60

DOI: 10.3310/hta23600

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

This Difference Elicitation in TriAls² (DELTA²) advice and recommendations document aims to help researchers choose the 'target difference' in a type of research study called a randomised controlled trial. The number of people needed to be involved in a study – the sample size – is usually based on a calculation aimed to ensure that the difference in benefit between treatments is likely to be detected. The calculation also accounts for the risk of a false-positive finding. No more patients than necessary should be involved.

Choosing a 'target difference' is an important step in calculating the sample size. The target difference is defined as the amount of difference in the participants' response to the treatments that we wish to detect. It is probably the most important piece of information used in the sample size calculation.

How we decide what the target difference should be depends on various factors. One key decision to make is how we should measure the benefits that treatments offer. For example, if we are evaluating a treatment for high blood pressure, the obvious thing to focus on would be blood pressure. We could then proceed to consider what an important difference in blood pressure between treatments would be, based on experts' views or evidence from previous research studies.

This document seeks to provide assistance to researchers on how to choose the target difference when designing a trial. It also provides advice to help them clearly present what was done and why, when writing up the study proposal or reporting the study's findings. The document is also intended to be read by those who decide whether or not a proposed study should be funded.

Clarifying a study's aim and getting a sensible sample size is important. It can affect not only those involved in the study, but also future patients who will receive treatment.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.819

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme or, commissioned/managed through the MRC–NIHR Methodology Research Programme (MRP), and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

This issue of the Health Technology Assessment journal series contains a project commissioned by the MRC–NIHR Methodology Research Programme (MRP). MRP aims to improve efficiency, quality and impact across the entire spectrum of biomedical and health-related research. In addition to the MRC and NIHR funding partners, MRP takes into account the needs of other stakeholders including the devolved administrations, industry R&D, and regulatory/advisory agencies and other public bodies. MRP supports investigator-led methodology research from across the UK that maximises benefits for researchers, patients and the general population – improving the methods available to ensure health research, decisions and policy are built on the best possible evidence.

To improve availability and uptake of methodological innovation, MRC and NIHR jointly supported a series of workshops to develop guidance in specified areas of methodological controversy or uncertainty (Methodology State-of-the-Art Workshop Programme). Workshops were commissioned by open calls for applications led by UK-based researchers. Workshop outputs are incorporated into this report, and MRC and NIHR endorse the methodological recommendations as state-of-the-art guidance at time of publication.

The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded under a MRC–NIHR partnership. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the MRC, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Cook et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk